Author's response to reviews

Title: What is the discrepancy of perception in primary care between physicians and patients in South Korea?: a qualitative study using focus group discussion

Authors:

Minsu Ock (ohohoms@naver.com)
Jung-Eun Kim (lydiasilver@gmail.com)
Min-Woo Jo (mdjominwoo@gmail.com)
Hyeon-Jeong Lee (trueprsc@hanmail.net)
Hyun Joo Kim (hyjkim2012@gmail.com)
Jin Yong Lee (jylee2000@gmail.com)

Version: 2
Date: 2 May 2014

Author's response to reviews:

As you mentioned, we put line numbers and page numbers in our manuscript. Also, tables were relocated in the page following reference.